Author's response to reviews

Title: Investment in online self-evaluation tests: A theoretical approach

Authors:

Francesco de Gara (f.degara@gmail.com)
William T Gallo (william.gallo@yale.edu)
Jonathan I Bisson (bissonji@cardiff.ac.uk)
Jerome Endrass (jerome.endrass@mail.com)
Stefan Vetter (stefan.vetter@access.uzh.ch)

Version: 2 Date: 11 April 2008

Author's response to reviews:

Dear Professor Ekkernkamp

As wished by your editorial production team we implemented the asked changes of our manuscript as following:

1. All qualifications and titles of the contributing authors were removed from the manuscript.
2. The table place holder was removed as well and it was controlled that table 1 is cited in the text.
3. The document was saved as .rtf (Rich Text Format) document and will be uploaded as such.

Sincerely yours,

Stefan Vetter, MD